## Vitamin D supplementation to prevent acute respiratory infections: individual participant data meta-analysis

Adrian R Martineau,<sup>1,2\*</sup> David A Jolliffe,<sup>1</sup> Lauren Greenberg,<sup>1</sup> John F Aloia,<sup>3</sup> Peter Bergman,<sup>4</sup> Gal Dubnov-Raz,<sup>5</sup> Susanna Esposito,<sup>6</sup> Davaasambuu Ganmaa,<sup>7</sup> Adit A Ginde,<sup>8</sup> Emma C Goodall,<sup>9</sup> Cameron C Grant,<sup>10</sup> Wim Janssens,<sup>11</sup> Megan E Jensen,<sup>12</sup> Conor P Kerley,<sup>13</sup> Ilkka Laaksi,<sup>14</sup> Semira Manaseki-Holland,<sup>15</sup> David Mauger,<sup>16</sup> David R Murdoch,<sup>17</sup> Rachel Neale,<sup>18</sup> Judy R Rees,<sup>19</sup> Steve Simpson Jr,<sup>20</sup> Iwona Stelmach,<sup>21</sup> Geeta Trilok Kumar,<sup>22</sup> Mitsuyoshi Urashima,<sup>23</sup> Carlos A Camargo Jr,<sup>24</sup> Christopher J Griffiths<sup>1,2,25</sup> and Richard L Hooper<sup>1</sup>

 <sup>1</sup>Centre for Primary Care and Public Health, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK
<sup>2</sup>Asthma UK Centre for Applied Research, Blizard Institute, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK
<sup>3</sup>Bone Mineral Research Center, Winthrop University Hospital, Mineola, NY, USA
<sup>4</sup>Department of Laboratory Medicine, Karolinska Institutet, Stockholm, Sweden
<sup>5</sup>Deptartment of Exercise, Lifestyle and Nutrition Clinic, Edmond and Lily Safra Children's Hospital, Tel Hashomer, Israel

- <sup>6</sup>Pediatric Highly Intensive Care Unit, Fondazione IRCCS Ca' Granda Ospedale Maggiore Policlinico, Università degli Studi di Milano, Milan, Italy
- <sup>7</sup>Department of Nutrition, Harvard School of Public Health, Boston, MA, USA
- <sup>8</sup>Department of Emergency Medicine, University of Colorado School of Medicine, Aurora, CO, USA
- <sup>9</sup>Department of Clinical Epidemiology and Biostatistics, McMaster University, Hamilton, ON, Canada
- <sup>10</sup>Department of Paediatrics: Child and Youth Health, Faculty of Medical and Health Sciences, University of Auckland, Auckland, New Zealand
- <sup>11</sup>Universitaire ziekenhuizen Leuven, Leuven, Belgium
- <sup>12</sup>Centre for Asthma and Respiratory Diseases, University of Newcastle, Newcastle, NSW, Australia
- <sup>13</sup>Dublin City University, Dublin, Ireland

- <sup>14</sup>Centre for Military Medicine, Finnish Defense Forces, University of Tampere, Tampere, Finland
- <sup>15</sup>Department of Public Health, Epidemiology and Biostatistics, Institute of Applied Health Sciences, College of Medical and Dental Sciences, University of Birmingham, Birmingham, UK
- <sup>16</sup>Department of Statistics, The Pennsylvania State University, Hershey, PA, USA
- <sup>17</sup>Department of Pathology, University of Otago, Christchurch, New Zealand
- <sup>18</sup>Queensland Institute of Medical Research Berghofer Medical Research Institute, Brisbane, QLD, Australia
- <sup>19</sup>Department of Epidemiology, Geisel School of Medicine at Dartmouth, Lebanon, NH, USA
- <sup>20</sup>Menzies Institute for Medical Research, University of Tasmania, Hobart, TAS, Australia
- <sup>21</sup>Department of Pediatrics and Allergy, Medical University of Łódź, Łódź, Poland <sup>22</sup>Institute of Home Economics, University of Delhi, New Delhi, India
- <sup>23</sup>Division of Molecular Epidemiology, Jikei University School of Medicine, Tokyo, Japan
- <sup>24</sup>Department of Emergency Medicine, Massachusetts General Hospital, Harvard Medical School, Boston, MA, USA
- <sup>25</sup>Medical Research Council and Asthma UK Centre in Allergic Mechanisms of Asthma, Blizard Institute, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, UK

\*Corresponding author a.martineau@qmul.ac.uk

Declared competing interests of authors: Susanna Esposito reports grants and personal fees from GlaxoSmithKline (GSK) plc (GSK House, Middlesex, UK), grants and personal fees from Pfizer Inc. (New York, NY, USA), grants and personal fees from Sanofi Pasteur MSD [Sanofi Pasteur (Lyon France) and Merck Sharp & Dohme Corp. (MSD, Kenilworth, NJ, USA)], grants from DuPage Medical Group (DMG, Downers Grove, IL, USA), personal fees from Valeas S.p.A. (Milan, Italy), and grants and personal fees from Vifor Pharma (Bern, Switzerland), outside the submitted work. Emma Goodall reports personal fees from GSK outside the submitted work. Wim Janssens reports grants from Instituut voor Innovatie door Wetenschap en Technologie (IWT)–Vlaanderen and from Laboratoires SMB (Brussels, Belgium) during the conduct of the study. David Mauger reports funding from the National Heart, Lung, and Blood Institute, MA, USA. Rachel Neale reports grants from the National Institutes of Health and the Medical Research Council during the conduct of the study. Judy R Rees reports that a use patent is held by Dartmouth College and Dr John A Baron for calcium as a chemopreventive agent. Dr Baron is not an author on this paper but is the principal investigator of the parent study from which the study by Rees (Rees JR, Hendricks K, Barry EL, Peacock JL, Mott LA, Sandler RS, et al. Vitamin D<sub>3</sub> supplementation and upper respiratory tract infections in a randomized, controlled trial. Clin Infect Dis 2013;57:1384–92) was conducted. The patent was previously licensed by Pfizer (with royalties), but has not been licensed for about 5 years. Judy R Rees is not involved in the patent and the patent does not involve vitamin D.

Published January 2019 DOI: 10.3310/hta23020

## **Plain English summary**

Vitamin D supplementation to prevent acute respiratory infections Health Technology Assessment 2019; Vol. 23: No. 2 DOI: 10.3310/hta23020

NIHR Journals Library www.journalslibrary.nihr.ac.uk

# **Plain English summary**

#### **Research question**

Does taking a vitamin D supplement help to prevent colds, flu and chest infections?

#### Background

Low blood levels of vitamin D (the 'sunshine vitamin') have been linked to an increased risk of colds, flu and chest infections, collectively termed 'acute respiratory infections' (ARIs). Clinical trials testing whether or not vitamin D supplements can prevent ARIs have had mixed results. The reason why vitamin D appears to work in some situations but not others is not understood. In order to answer this question, we obtained data from individuals who took part in previous clinical trials, combined them and analysed them to answer two questions:

- 1. Does vitamin D reduce the overall risk of ARIs, broadly defined?
- 2. Do some people benefit more from taking vitamin D than others?

### **Included studies**

We obtained raw data on a total of 10,933 people from 25 trials conducted in 15 countries. Participants were aged from 0 to 95 years. All of the studies compared vitamin D with placebo (dummy medication), which is the gold standard trial design.

#### **Key results**

Overall, vitamin D supplements reduced the risk of having at least one ARI from 42% to 39%. We also showed that vitamin D had greater protective effects when it was given daily or weekly to people with the lowest vitamin D levels: the risk of having at least one ARI was reduced from 60% to 32% in these individuals. Vitamin D was not effective in protecting against ARIs when it was given in large, widely spaced doses. Taking vitamin D supplements was found to be safe.

### Conclusion

Taking a vitamin D supplement can protect against ARIs. The strongest effects are seen when a daily or weekly supplement is given to people with the lowest vitamin D levels.

## **Health Technology Assessment**

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.236

Health Technology Assessment is indexed in MEDLINE, CINAHL, EMBASE, The Cochrane Library and the Clarivate Analytics Science Citation Index.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

#### Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

#### **HTA programme**

The HTA programme, part of the National Institute for Health Research (NIHR), was set up in 1993. It produces high-quality research information on the effectiveness, costs and broader impact of health technologies for those who use, manage and provide care in the NHS. 'Health technologies' are broadly defined as all interventions used to promote health, prevent and treat disease, and improve rehabilitation and long-term care.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

For more information about the HTA programme please visit the website: http://www.nets.nihr.ac.uk/programmes/hta

#### This report

The research reported in this issue of the journal was funded by the HTA programme as project number 13/03/25. The contractual start date was in October 2014. The draft report began editorial review in January 2017 and was accepted for publication in October 2017. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HTA programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Martineau *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

### **NIHR Journals Library Editor-in-Chief**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

#### **NIHR Journals Library Editors**

**Professor Ken Stein** Chair of HTA and EME Editorial Board and Professor of Public Health, University of Exeter Medical School, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals)

**Professor Matthias Beck** Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Scientific Advisor, NETSCC, UK

**Dr Catriona McDaid** Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK

**Professor James Raftery** Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

**Professor Helen Snooks** Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

**Professor Jim Thornton** Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk